Author's response to reviews

Title: The effect of ethnicity on prescribing practice and treatment outcome in inpatients suffering from schizophrenia in Greece

Authors:

Athanassios Douzenis (thandouz@med.uoa.gr)
Athanassios Apostolopoulos (th.apostolopulos@gmail.com)
Dionisios Seretis (seretisdion@hotmail.com)
Emmanouil N Rizos (erizos@med.uoa.gr)
Christos Christodoulou (christo.christodoulou@gmail.com)
Lefteris Lykouras (panpsyclin@attikonhospital.gr)

Version: 2 Date: 20 October 2010

Author's response to reviews:

Following our correspondence please accept our revised submission which specifies that ethical approval for this research was granted by the Attikon University General Hospital ethics committee and also the scientific committee